Author's response to reviews

Title: Modified endoscopic submucosal dissection with enucleation for treatment of gastric subepithelial tumors originating from the muscularis propria layer

Authors:

Yin-Yi Chu (chu2235@yahoo.com)
Jau-Min Lien (lienoski@seed.net.tw)
Ming-Hung Tsai (mhtsai@adm.cgmh.org.tw)
Cheng-Tang Chiu (ctchiu@adm.cgmh.org.tw)
Tse-Ching Chen (ctc323@adm.cgmh.org.tw)
Kuo-Ching Yang (M001003@ms.skh.org.tw)
Soh-Ching Ng (angelang@adm.cgmh.org.tw)

Version: 3 Date: 28 August 2012

Author's response to reviews: see over
Dear editor and reviewers,

This manuscript has been edited. We have removed all the words of en bloc from the manuscript as your suggestion.

Thank you for your comment.

Yin-Yi Chu, M.D.